Author's response to reviews

Title: The associated expression of Maspin and Bax proteins as a potential prognostic factor in intrahepatic cholangiocarcinoma

Authors:

Antonello A. Romani (antonello.romani@studenti.unipr.it)
Paolo Soliani (paolo.soliani@unipr.it)
Silvia Desenzani (silvia.desenzani@unipr.it)
Angelo F. Borghetti (angelo.borghetti@unipr.it)
Pellegrino Crafa (RCrafa@ao.pr.it)

Version: 3 Date: 23 October 2006

Author's response to reviews: see over
We herewith resubmit our revised manuscript entitled, “The associated expression of Maspin and Bax proteins as a potential prognostic factor in intrahepatic cholangiocarcinoma” (MS: 3298159231066178) by Antonello A. Romani, Paolo Soliani, Silvia Desenzani, Angelo F. Borghetti and Pellegrino Crafa for publication in BioMed Central Cancer.

We have incorporated all of the editor requirements.

1. As recommended the text was proofed for typos throughout the manuscript by a native English speaking colleague who at present works as professional copyediting.

2. Formatted the text in accordance with BMC style.

3. The manuscript was rewritten in those places where language was less than optimal.

Yours sincerely,

Angelo F. Borghetti
Antonello A. Romani